Author’s response to reviews

Title: Use of Natural language processing to improve predictive models for imaging utilization in children presenting to the Emergency Department

Authors:

Xingyu Zhang (zhangxyu@umich.edu)
Fernanda Bellolio (Bellolio.Fernanda@mayo.edu)
Pau Medrano-Gracia (p.medrano@auckland.ac.nz)
Konrad Werys (konrad.werys@cardiov.ox.ac.uk)
Sheng Yang (yasheng@umich.edu)
Prashant Mahajan (pmahajan@med.umich.edu)

Version: 2 Date: 03 Sep 2019

Author’s response to reviews:

We thank the reviewers for their generous comments on the manuscript, and have edited the manuscript to address their concerns. We included all the responses to the comments in a separate document and uploaded it.